,



Geophysical Research Letters

RESEARCH LETTER

10.1029/2020GL091771

Key Points:

- Hierarchical analysis of marine particle data identifies 0.63 ± 0.06 overall size-sinking scaling
- Open ocean and *in situ* measurements have weaker scaling than coastal/laboratory/*ex situ* measurements
- Results suggest importance of small particles for flux and how to harmonize biogeochemical models

Supporting Information:

- Table S1
- Table S2
- Dataset S1Dataset S2
- Dataset S2

Correspondence to:

B. B. Cael, cael@noc.ac.uk

Citation:

Cael, B. B., Cavan, E. L., & Britten, G. L. (2021). Reconciling the sizedependence of marine particle sinking speed. *Geophysical Research Letters*, 48, e2020GL091771. https://doi. org/10.1029/2020GL091771

Received 20 NOV 2020 Accepted 22 DEC 2020

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Reconciling the Size-Dependence of Marine Particle Sinking Speed

B. B. Cael¹, Emma L. Cavan², and Gregory L. Britten³

¹National Oceanography Centre, Southampton, UK, ²Imperial College London, UK, ³Program in Atmospheres, Oceans, and Climate, Massachusetts Institute of Technology, Cambridge, MA, USA

Abstract Sinking particles are critical to the ocean's "biological pump," sequestering carbon from the atmosphere. Particles' sinking speeds are a primary factor determining fluxes and subsequent ecological and climatic impacts. While size is a key determinant of particles' sinking speeds, observations suggest a variable size-sinking relationship, affected by other particle properties, resulting in substantial spread in parameterizations of particle sinking and fluxes. We compile particle size-sinking observations and apply hierarchical Bayesian statistical models to resolve the size-sinking relationship while accounting for other factors. We find an overall scaling close to the general Navier-Stokes drag equation, and differences between particle types, open ocean versus coastal/laboratory particles, and in situ versus ex situ methods. These results can help harmonize how Earth system models parameterize particle fluxes and support a weaker size-dependence than often assumed, with implications for the flux contribution of small particles and the predicted future shrinking of marine particle populations.

Plain Language Summary Sinking particles in the ocean comprise a major flux within the global carbon, nutrient, and oxygen cycles. Particle flux is strongly influenced by sinking speed, which in turn is thought to be strongly influenced by particle size. However, observed variability in particle geometry and composition complicates this size-sinking relationship and has introduced significant uncertainty into ocean flux models. Using hierarchical Bayesian statistical modeling, we show marine sinking particle observations are broadly consistent with the general Stokes' theory for particle terminal speeds. These results revise flux models currently used in earth system model predictions and have implications for the predicted shrinking of marine particles with climate change.

1. Introduction

The continual flux of organic carbon from the surface to the deep ocean sequesters carbon from exchanging with the atmosphere, and thus maintains atmospheric CO_2 and global temperature appreciably below what they would otherwise be (Sarmiento & Gruber, 2006; Sigman & Boyle, 2000; Sigman et al., 2010; Williams & Follows, 2011). The majority of organic carbon flux occurs via the gravitationally forced sinking of marine particles created in the sunlit upper ocean from photosynthesis. As they sink, particles are decomposed ("remineralized") which releases inorganic carbon back to solution in seawater at the depth where remineralization occurs (Herndl & Reinthaler, 2013). The balance of sinking speed and remineralization thus determines the depth of carbon sequestration, which in turn determines the residence time before the sequestered carbon can be released back to the atmosphere (Buesseler & Boyd, 2009; DeVries et al., 2012; DeVries & Primeau, 2011). Particulate fluxes are proportional to particles' sinking speeds, so for a given rate of remineralization, faster sinking speeds lead to deeper remineralization and lower atmospheric CO_2 (Kwon et al., 2009). Thus, understanding the controls on particle sinking speeds is critical for understanding the biological pump's role in marine biogeochemistry and climate.

Physical theory (Clift et al., 2005; Stokes et al., 1851) suggests that a particle's sinking speed w (m/d) should be controlled by its excess density $\Delta \rho$ (kg/m³) relative to the density of the surrounding fluid ρ_o (kg/m³), its shape, and its size, commonly expressed in terms of equivalent spherical diameter d (mm). For an ideal solid sphere at low Reynolds number in a quiescent fluid, this speed is given by Stokes' Law:

N

$$v = \frac{g}{18\mu}\Delta\rho \ d^2,\tag{1}$$



Geophysical Research Letters



Figure 1. Left: Particle size (expressed as equivalent spherical diameter *d* (mm)) versus sinking (speed *w* (m/d)) from the sources in Table S2 and from models. Individual studies are randomly colored, with ordinary least squares linear regressions of log-transformed data overlaid. The solid orange line is the median result of the hierarchical regression to the overall data set ($\log_{10}(\alpha) = 2.11, \beta = 0.63$) and the solid black line is the result of an ordinary least squares regression ($\log_{10}(\alpha) = 2.11, \beta = 0.45$). Dashed, dotted, and dash-dotted lines are common model parametrizations of size-sinking relationships (Dunne et al., 2010; Dutkiewicz et al., 2015; Gehlen et al., 2006; Niemeyer et al., 2019; Yool et al., 2013). Thin black line is a Stokes'-Law-like $\beta = 2$. Right: 100-m normalized fluxes (flux at a given depth divided by flux at 100 m) for particle populations of different sizes with a remineralization rate of k = 0.24 days⁻¹ (Collins et al., 2015) and sinking speeds from relationships in the left panel ($\beta = 2$ not shown).

where g is the gravitational constant and μ is the dynamic viscosity of seawater (kg/ms). This simple equation suggests that sinking speeds scale positively and nonlinearly with size and is practically appealing because of its simplicity and basic theoretical underpinnings. However, it has been noted that observations of sinking particles typically deviate widely from the basic relationship (Diercks & Asper, 1997; Laurenceau-Cornec et al., 2015, 2019; Figure 1). Individual studies of different assemblages of particles over the past few decades find size-sinking relationships ranging from a Stokes'-Law-like relationship (Azetsu-Scott & Johnson, 1992) to an overall decrease of sinking velocity with size (McDonnell & Buesseler, 2010). The diversity of assumed relationship demonstrates the importance of various factors that obscure the overall size-dependency of sinking speeds such as inclusion of ballast minerals (Klaas & Archer, 2002) or "sticky" transparent exopolymers (Engel & Schartau, 1999), ambient temperature (Iversen & Ploug, 2013), or particles' geometry (Johnson et al., 1996). Several variations to Equation 1 have been proposed for nonspherical and porous particles, generally with scaling exponents <2 (Tang & Raper, 2002), but it is not clear whether any of these is appropriate for the large diversity of natural marine particles.

As a result, Earth system models use a variety of different size-sinking relationships to parameterize particulate organic matter fluxes (e.g., Dunne et al., 2010; Dutkiewicz et al., 2015; Gehlen et al., 2006; Kriest & Oschlies, 2008; Yool et al., 2013; Figure 1). These size-sinking relationships yield order-of-magnitude differences in sinking speeds for particles of the same size (Figure 1) and therefore predict widely different fluxes and introduce uncertainty into models of ocean biogeochemistry (Gehlen et al., 2006; Niemeyer et al., 2019). Furthermore, size-sinking relationships have additional implications for climate change projections, as plankton models suggest a shrinking of cell size with warming, which would positively feedback onto atmospheric CO_2 due to reduced particle sinking speeds predicted from size-sinking relationships (Bopp et al., 2005; Finkel et al., 2010). Though it has become increasingly clear that environmental context, particle type, and other factors can limit our ability to extrapolate observed empirical relationships to estimate large scale fluxes (Laurenceau-Cornec et al., 2015), these uncertainties have not yet been treated in a statistical fashion, which could help harmonize biological pump parameterizations in ocean models. Determining particles' size-sinking relationships is not only of interest for global modeling but also for the increasingly prevalent technique of estimating particle fluxes from optically derived ambient particle size distributions (Guidi et al., 2008) using, for example, the Underwater Vision Profiler (Picheral et al., 2010). These methods require knowledge of how particle sinking speed and mass scale with size, and how these might vary for different particles or environments.

Here, we attempt to resolve the size-sinking relationship in marine particles by assembling the largest data set of particle size and sinking speed to date. Our objective is to quantify size-sinking scaling relationships from particle observations, provide confidence bounds on this scaling, and its dependence on other factors. We develop a Bayesian hierarchical model framework, motivated from a generalized Navier-Stokes drag equation, to estimate a universal size-scaling relationship while taking into account context-dependent factors. For nonspherical, heterogeneous particles like "marine snow" aggregates or zooplankton fecal pellets, a more general Navier-Stokes drag equation can be used to describe sinking speed (Laurenceau-Cornec et al., 2019; Stokes et al., 1851):

$$w = \left(\frac{4g}{3\rho_o}\frac{\Delta\rho}{C_D}d\right)^{1/2},\tag{2}$$

where C_D is the dimensionless drag coefficient. Here both $\Delta \rho$ and C_D are size-dependent. The former dependence is typically expressed in terms of the fractal dimension f as $\Delta \rho \propto d^{f-3}$, whereas C_D scales inversely with Reynolds number for solid spheres. (f can be thought of in less technical terms as quantitatively expressing how much less volume-filling larger particles are than smaller particles.) In general, f and C_D are not well-characterized for complex natural particle shapes and structures (Loth, 2008; Tang & Raper, 2002), though both $\Delta \rho$ and C_D should decrease with d. Thus, according to Equation 2 sinking speed should scale approximately with the square root of particle size diameter, but in potentially complex or variable ways, depending on factors that modulate the parameters of the generalized drag equation in Equation 2.

A general parameterization of Equation 2 has the form

$$w(d) = \alpha d^{\beta} \tag{3}$$

where β is the dimensionless scaling exponent and α is the prefactor coefficient (m/d), or the speed of a 1 mm particle. Viewed this way, α and β are unknown parameters with variability that depends on particle type and environmental factors. We formulate Equation 3 as a hierarchical Bayesian model and allow deviations from an overall size-scaling relationship according to: (1) aggregate particles versus. fecal pellets, (2) coastal versus open ocean versus laboratory measurements, (3) *in situ* versus. indirect *in situ* versus flow chamber versus other measurement methodologies, (4) "warm" (\geq 14°C) versus "cold" (<14°C) environments, (5) diatom versus nondiatom for aggregate particles, and (6) ballast versus nonballast for aggregate particles (see Methods for more detail). Hierarchical Bayesian regression has enjoyed success in other scientific fields (Gelman & Hill, 2006), but has not previously been applied to marine particle datasets. We fit the model to the compiled data set, representing sinking speeds from a wide variety of contexts, and report posterior probability distributions for the overall scaling and context-specific deviations. We discuss the implications of our estimated size-sinking relationships for biogeochemical models and for the global carbon cycle.

1.1. Overall Scaling

The hierarchical model suggests a posterior median overall scaling exponent $\beta = 0.63$ (with a standard deviation ± 0.06) and a median coefficient $\alpha = 129 \pm 26$ m/d. The overall and study-specific posterior distributions for α and β are plotted in Figure 3. Despite large study-to-study variability in both parameters, the hierarchical model finds an overall scaling exponent much smaller than Equation 1 ($\beta = 2$) and roughly in line with Equation 2 ($\beta = 1/2$). The difference between the 1/2-scaling in Equation 2 and the $\beta = 0.63 \pm 0.06$ scaling in Figure 3 may be due to environmental and/or methodological discrepancies (see below) but in general suggests that the ratio $\Delta \rho/C_D$ changes only weakly with particle size. We note that the overall parameter distributions are virtually unchanged across all the different model networks we used, indicating that these overall α and β parameter values are robust (top level in Figure 2). The hierarchical model also finds a higher β than the $\beta = 0.45 \pm 0.01$ found by an ordinary least squares





Figure 2. Schematic of hierarchical regression. The likelihood of individual observations is evaluated conditional on the hierarchy of parameters defined upstream. Individual studies' scaling coefficients are drawn from groups (i.e., dividing studies by type of particle, type of environment, type of method, or temperature – or no grouping is imposed in the three-level network case) of studies' distributions, which in turn are drawn from the overall scaling coefficients' distribution. Here *p* refers to a probability density function, *i* is an index indicating different groups, *j* is an index indicating different studies within groups, and *k* is an index indicating different individual particle observations within studies, that is, w_{ijk} is the *k*th sinking speed observation given the parameters' values is evaluated on the "data" level. See Materials and Methods for details.

(OLS) regression of the total aggregated data set (a posterior probability of p > 99%), but almost exactly the same α , suggesting faster sinking velocities than OLS for particles >1 mm and slower sinking velocities for particles <1 mm (Figure 1).

The total flux ${\cal F}(g~m^{-2}~d^{-1})$ of an assemblage of particles can be expressed (Guidi et al., 2008) as

$$\mathcal{F} = \int_0^\infty N(d) w(d) m(d) \, \mathrm{d}d \tag{4}$$

where N(d) is the number of particles of diameter d, w(d) is the average size-sinking speed relationship, and m(d) is the average size-mass relationship. Thus substituting Equation 3 into Equation 4 shows that total particle flux \mathcal{F} is proportional to the coefficient α , and that the relative contribution of small and large particles to \mathcal{F} will be determined by β and how both N and m scale with d. The utility of the hierarchical approach is primarily in constraining β , as seen by the fact that the overall $\alpha = 129 \pm 26$ m/d matches both the overall α found by a simple OLS regression and a common "rule of thumb" estimate that particles sink on the order of 100 m/d. An overall scaling exponent of $\beta = 0.63 \pm 0.06$ suggests a much more substantial contribution of small particles to total flux than if particles sank like solid spheres or individual cells (Laws, 1975; Niemeyer et al., 2019), as has been increasingly recognized (Alonso-González et al., 2010; Durkin et al., 2015; Riley et al., 2012). This is further supported by the fact that a low β is consistent with a low fractal dimension f, that is, a low mass-size scaling exponent, even if the quantitative link between β and f is uncertain (Johnson et al., 1996; Tang & Raper, 2002). Furthermore, this suggests particle fragmentation is less effective at attenuating total flux; if a particle becomes fragmented, the change in average sinking speed of the resulting (smaller) particle fragments will be less than if β were larger. A $\beta = 0.63 \pm 0.06$ also suggests that future decreases in particle size distributions (Bopp et al., 2005; Finkel et al., 2010) would not change the total flux \mathcal{F} nearly as much as if β were more akin to the larger values for individual cells (Laws, 1975) or solid objects (Clift et al., 2005; Stokes et al., 1851).

The differences in relative contribution to total flux of small and large particles between our overall result and example model parameterizations can be seen in Figure 1. For a parameterization using $\beta = 1.17$, effectively all of the flux below the upper few hundred meters of the water column is due to very large (~3 mm) particles, as smaller particles are attenuated rapidly. In contrast, a step-function parameterization vastly overpredicts flux due to ~300 μ m particles, and vastly underpredicts flux due to ~30 μ m particles, compared to the hierarchical model's estimated relationship. Our hierarchical β of 0.63 ± 0.06 is both globally plausible and biogeochemically relevant as supported by a recent study (Niemeyer et al., 2019) which showed that a $\beta = 0.62$ appreciably outperforms a $\beta = 1.17$ in a global biogeochemical model's ability to simulate tracer distributions. The value of 0.62 in that study was derived from one of the equations that have been proposed to link β to particles' fractal dimension *f* (Kriest, 2002).

1.2. Group-Level Scalings

Our hierarchical models' group-level results largely confirm general expectations about the deviations from the overall size-sinking relationship (Figure 4; Table S1). First, both α and β are larger (p > 99% and $p \approx 86\%$, respectively) for fecal pellets than for aggregates. The higher α is expected as fecal pellets are generally thought to be denser (meaning larger $\Delta \rho$) and less rough than aggregates (meaning smaller C_D ; Ebersbach et al., 2011; Yoon et al., 2001). The higher β is also partly explained by lower porosity, or higher fractal dimension, of the fecal pellets, as the effective density of fecal pellets should decrease comparatively less with





Figure 3. Posterior distributions for α and β for the three-level hierarchical model (i.e., no groupings imposed). Thick black lines are the posteriors for the overall parameters (top level in Figure 2) with medians and standard deviations given in the legends; colored lines are the posteriors for the individual studies ("study" level in Figure 2).

increasing particle size, resulting in a higher β . In any case these results provide supporting evidence that aggregates' and fecal pellets' sinking should be parameterized differently in biogeochemical models (Table S1).

We find larger α in warm environments ($p \approx 76\%$), expected due to the lower viscosity of seawater at higher temperatures (Taucher et al., 2014) resulting in faster sinking speeds for all particles. α is also higher for ballasted ($p \approx 94\%$) and diatom-enriched ($p \approx 75\%$) aggregates, as expected from the higher densities of these materials (Engel et al., 2009). Diatom-enriched and ballasted aggregates are not only expected to be composed of more dense material but also more compact (Laurenceau-Cornec et al., 2019), and therefore expected to have a higher β , also borne out by these analyses ($p \approx 75\%$ and $p \approx 69\%$ respectively). It is unclear whether β would be expected to be higher, lower, or the same for particles from warmer environments; β is found to be larger for the warm group ($p \approx 72\%$), further indicating faster sinking speeds for these particles.

We find larger α ($p \approx 91\%$) and β ($p \approx 77\%$) for coastal particles compared to open ocean particles. Particles from coastal environments are expected to sink faster because coastal environments are subject to loading of minerals and other dense material from rivers, sediment resuspension, dust deposition, and other sources, and also because coastal ecosystems tend to be more dominated by diatoms (Armbrust, 2009). Similarly, as laboratory-generated particles are typically generated in volumes with artificially high collision rates (Jackson, 2015), from diatom cultures or from phytoplankton collected in close proximity to the laboratory, they should also have a larger α and β for the same reasons as above, which we also find ($p \approx 85\%$ and $p \approx 83\%$, respectively). Though these differences are unsurprising, they are important because much of our understanding and parameterization of open ocean particles comes from coastal and laboratory studies, which may influence not only the magnitude of sinking speeds (α) but also their size dependence (β). We find a β of 0.56 \pm 0.09 for the open ocean group, that is, within one standard deviation of the 1/2scaling from Equation 2. When modeling sinking speeds of open ocean particles, we therefore recommend the $\beta = 0.56 \pm 0.09$ exponent. As above, we interpret the lower open ocean β to result from higher porosity or lower fractal dimension of open-ocean particles.

Finally, surprising and potentially illuminating differences are revealed by the hierarchical model which groups by methodology. The *in situ* direct methods (those which measure sinking speeds, e.g., Carder et al., 1982) and indirect methods (those which infer speeds from the ratio of flux to concentration, Mc-Donnell & Buesseler, 2010) are in very good agreement in terms of α and β . The indirect group's parameters have larger uncertainties, likely due to smaller number of studies using these methods, but their central estimates are indistinguishable. In contrast, the "flow chamber" (Ploug & Jørgensen, 1999) and "other ex situ methods" (e.g., roller tank (Shanks & Edmondson, 1989) or settling column (Smayda, 1969)) groups have substantially different posterior distributions for both parameters. Both have larger α values than the "in *situ*" group ($p \approx 98\%$ and p > 99%, respectively), but also larger β values ($p \approx 93\%$ and $p \approx 75\%$, respectively). As with the open ocean group above, the in situ direct and indirect methods groups are consistent with the 1/2 scaling from Equation 2. While the differences between methods distributions may be in part explained by the types of particles considered in the studies employing these methods, these ex situ methods have been employed widely and in a variety of contexts, and the evidence for these differences is strong. Fragile particles that would be disturbed by collection and therefore not easily captured by exsitu techniques (Giering et al., 2020) should be expected to sink slower and be less compact (i.e., have lower fractal dimensions), accounting for these parameter differences, but these results do suggest that such particles are a non-negligible component of in situ sinking particle populations.





Figure 4. Posterior distributions for α and β for the group levels of hierarchical models ("group" level in Figure 2). Vertical lines show the medians of each distribution. n. b. the bottom two rows do not include fecal pellet data, and legends from the left column apply to the right column. See Table S1 for medians and confidence intervals.

2. Discussion

These results help resolve the size-sinking relationship of marine particles, with implications for particulate carbon fluxes and our understanding of the global carbon-climate system. Our approach establishes a baseline for a universal size-sinking relationship and quantifies important context dependent deviations. The hierarchical framework allowed us to account for unbalanced sampling across different contexts and provides a rigorous uncertainty bound that can also be propagated through subsequent carbon flux analyses. With this new hierarchical analysis across datasets, we have provided evidence that size-sinking relationships hold at broad scales and under various environmental conditions, and have quantified them for use in biogeochemical modeling and optical flux estimation.

These parameterizations can inform not only size-spectrum-resolving biogeochemical models (Kriest & Oschlies, 2008) but also those with only a few (e.g., two) size classes (Yool et al., 2013), as well as those with implicit flux profiles (Cael & Bisson, 2018; Najjar et al., 2007). The parameterizations are also important for the rapidly expanding field of optically derived particle flux estimation (Giering et al., 2020). It also has important implications for particle fluid dynamics that the study-specific scaling exponent svary greatly from study to study while the central scaling exponent is close to that of the generalized Navier-Stokes' drag equation exponent of 1/2 (Equation 2). This suggests that the size-dependencies of excess densities and drag coefficients of natural aggregates vary between particle populations, but that overall they are in approximate balance.

To date, carbon flux models generally combine sinking speed and remineralization relationships to achieve a predicted distribution of biogeochemical tracers that approximates observations (Najjar et al., 2007). However, the variety of size-sinking relationships in use (Figure 1) suggests that the two processes are being combined in highly uncertain ways. Uncertainty in size-sinking and remineralization relationships have direct implications for climate change (Völker et al., 2016). Warmer oceans are predicted to enhance water column stratification and reduce nutrient fluxes, which is predicted to shift cells toward smaller sizes (Finkel et al., 2010). Shifts in average cell size would lead to decreases in sinking speed and therefore carbon flux, which feedbacks onto atmospheric CO₂ and global climate. Our results suggest a comparatively weak size-dependence of sinking speed, which suggests this climate feedback may be weaker than previously assumed. A better parameterization of the size-sinking relationships should also allow better isolation in Earth System Models of remineralization relationships that are expected to vary

with warming due to faster rates of heterotrophic metabolism (Cael & Follows, 2016; Cavan & Boyd, 2018; Cavan et al., 2019), especially given the large vertical gradients in temperature in the upper ocean. Improved parameterization of particle flux thereby provides a means to better predict changes in the ocean carbon cycle with climate change.

Our results underscore the importance of not only size but also composition, structure, and/or density in determining particle sinking speed; as seen in Figure 1, for example, a small aggregate can sink at similar velocity as that of a large aggregate. While much remains unknown or uncertain about the controlling factors for sinking speeds of *in situ* formed aggregates, comparatively much more is known about the plankton communities from which these aggregates are formed, including where and when different types of plankton are prevalent or dominant (Barton et al., 2013). Therefore, the distinct parameterizations we provide for fecal pellets, ballasted aggregates, and nonballasted aggregates (Figure 4, Table S1) that extract information

about the size-sinking relationships for different particle/aggregate types may be particularly useful for improving particle flux estimates based on community composition. This is an especially appealing possibility as it is becoming possible to differentiate community structure information (e.g., whether diatoms vs. coccolithophores vs. flagellates are dominant) from remote sensing (Kramer & Siegel, 2019), though challenges persist for the rigorous application of such approaches (Cael et al., 2020). Given suitable data, it would be particularly valuable in this context to estimate scaling relationships for aggregates formed from plankton communities dominated by different functional groups such as diatoms versus coccolithophores.

While our results help bridge the gap between modeled fluxes and observed sinking speed variability, more work remains to be done in understanding the unexplained variance (Figure 1). The statistical framework herein is useful for propagating the uncertainty due to explained variance in the data. Some of the unexplained variance is likely due to spatial sampling variability and measurement error, but other more mechanistic relationships should also be developed to explain the variance that may provide more predictive relationships for modeling sinking particle fluxes. Note also that the relationships we find are bulk scalings that are of interest for predicting particles' impact on biogeochemical fluxes; individual particles' sinking speeds will undoubtedly vary around these scalings.

In summary, our results advance our understanding of ocean carbon fluxes through an improved, data-driven description of the gravitationally driven sinking of marine particles. Our data compilation and hierarchical modeling suggests an approximately 0.6-scaling of sinking speed with particle size, with implications for the role of small particles in the biological pump and for how carbon flux will vary with climate change. Despite these results, the data compilation also points to significant unexplained variability that requires further study to confidently predict changes in these climatically important ocean processes.

3. Materials and Methods

3.1. Size-Sinking Data Compilation

We compiled observations of 5,655 particles' sizes and sinking speeds from 54 studies published in 32 papers (Alldredge & Gotschalk, 1988, 1989; Azetsu-Scott & Johnson, 1992; Belcher et al., 2016a, 2016b; Carder et al., 1982; Cavan et al., 2018; Deibel, 1990; Diercks & Asper, 1997; Engel & Schartau, 1999; Gibbs, 1985; Engel et al., 2009; Hawley, 1982; Hill et al., 1998; Iversen & Ploug, 2010, 2013; Iversen & Robert, 2015; Iversen et al., 2010, 2017; Jouandet et al., 2011; Kajihara, 1971; Laurenceau-Cornec et al., 2015, 2019; McDonnell & Buesseler, 2010; Nowald et al., 2009; Shanks & Trent, 1980; Smayda, 1969; Syvitski et al., 1995; Van der Jagt et al., 2018); see Table S2. Size data were standardized to equivalent spherical diameter in units of millimeters. Sinking data were standardized to sinking speed in units of meters per day. Data that we were otherwise not able to access were digitized (https://automeris.io/WebPlotDigitizer/). Previous analyses of digitized data suggest that the error introduced by digitization is negligible (<1%; Cael & Bisson, 2018). Particles were classified as aggregate or fecal pellet, diatom-enriched or not, ballasted or not, and by their environment according to information from the publications. Temperatures were classified as "warm" if ≥14°C and "cold" otherwise. We chose this coarse temperature classification because many of the publications did not report environmental temperature, in which cases we supplemented by referencing the sampling locations and times to a monthly climatology (Reynolds et al., 2007); 14°C is the median of that climatology. Methods were classified into in situ direct observations, indirect in situ observations (i.e., estimation by the ratio of flux to concentration, e.g., Jouandet et al., 2011), flow chamber observations (Ploug & Jørgensen, 1999), and other ex situ methods (which measure speed by distance traveled per unit time, e.g., settling column (Smayda, 1969)). See Table S2 for the sources and classifications of the data.

3.2. Bayesian Hierarchical Analysis

We consider the model that individual observations vary according to study-specific size-sinking scaling relationships. These study-specific scalings then deviate from an overall scaling relationship for marine sinking particles. We also consider models where the study-specific scalings deviate from group-specific scaling relationships, which in turn deviate from an overall scaling relationship. We formalize each of these by



fitting a hierarchical model to these data, which captures group, within-group, and within-study variance in a single statistical framework. Each hierarchical model can be represented as a network of conditional probabilities (see Figure 2). At the "overall" level we assume that the overall particle size-sinking relationship can be represented as a scaling relationship (Equation 3) governed by overall parameters $\log_{10} \alpha$ and $\overline{\beta}$. Group level parameters (indexed by *i*) then deviate from these overall distributions according to the group parameter distributions

$$p(\overline{\log_{10}\alpha}_{i} | \overline{\log_{10}\alpha}) \sim \mathcal{N}(\overline{\log_{10}\alpha}, \sigma_{i}^{\alpha}),$$

$$p(\overline{\beta}_{i} | \overline{\beta}) \sim \mathcal{N}(\overline{\beta}, \sigma_{i}^{\beta}).$$

There is no group level in the case where studies are not partitioned into groups. Similarly, we assume that at the study level (indexed by *j*) the studies' mean parameters deviate from these group distributions according to the study distributions

$$p(\log_{10}\alpha_{ij} \mid \overline{\log_{10}\alpha_i}) \sim \mathcal{N}(\overline{\log_{10}\alpha_i}, \sigma_{ij}^{\alpha}),$$
$$p(\beta_{ii} \mid \overline{\beta_i}) \sim \mathcal{N}(\overline{\beta_i}, \sigma_{ij}^{\beta}).$$

Lastly, the observation level of analysis assumes Gaussian errors on the logarithms of individual measurements (indexed by k) of (w, d),

$$p(\log_{10}w_{ijk} \mid \log_{10}\alpha_{ij}, \beta_{ij}) \sim \mathcal{N}(\log_{10}\alpha_{ij} + \beta_{ij} \times \log_{10}d_{ijk}, \sigma^{w}).$$

We assume broad uniform priors over all α , β , and σ parameters. Broad bounds were imposed to keep the parameters from nonphysical values. No posteriors were close to the bounds and so these bounds did not impact the inference.

Models are fit via Markov Chain Monte Carlo (MCMC) sampling, drawing random samples of model parameters and propagating them through the hierarchical network. Doing so many times yields a set of samples from the posterior distribution, that is, the distribution of the model parameters after the data are taken into account. We use the RStan and brms software package to perform the MCMC sampling (Betancourt et al., 2017; Bürkner et al., 2017; Carpenter et al., 2017). We run 4 chains with 5,000 iterations per chain. We discarded the first 2,500 samples from each chain as warmup, leaving a total of 10,000 samples from each posterior. All \hat{R} statistics for the MCMC chains were between 1.0 and 1.1 which indicated convergent mixing.

Data Availability Statement

All code and data will be made shared on GitHub and archived on Zenodo, and the data will also be archived on Pangaea, should this manuscript be accepted for publication. The data are also available via the supporting information.

References

- Alldredge, A. L., & Gotschalk, C. (1988). In situ settling behavior of marine snow 1. Limnology & Oceanography, 33(3), 339–351. https://doi.org/10.4319/lo.1988.33.3.0339
- Alldredge, A. L., & Gotschalk, C. (1989). Direct observations of the mass flocculation of diatom blooms: Characteristics, settling velocities and formation of diatom aggregates. Deep Sea Research Part A. Oceanographic Research Papers, 36(2), 159–171. https://doi. org/10.1016/0198-0149(89)90131-3
- Alonso-González, I. J., Arístegui, J., Lee, C., Sanchez-Vidal, A., Calafat, A., Fabrés, J., et al. (2010). Role of slowly settling particles in the ocean carbon cycle. *Geophysical Research Letters*, 37(13). https://doi.org/10.1029/2010GL043827
- Armbrust, E. V. (2009). The life of diatoms in the world's oceans. Nature, 459(7244), 185–192. https://doi.org/10.1038/nature08057
 - Azetsu-Scott, K., & Johnson, B. D. (1992). Measuring physical characteristics of particles: A new method of simultaneous measurement for size, settling velocity and density of constituent matter. *Deep Sea Research Part A. Oceanographic Research Papers*, 39(6), 1057–1066. https://doi.org/10.1016/0198-0149(92)90039-V
 - Barton, A. D., Pershing, A. J., Litchman, E., Record, N. R., Edwards, K. F., Finkel, Z. V., et al. (2013). The biogeography of marine plankton traits. *Ecology Letters*, *16*(4), 522–534. https://doi.org/10.1111/ele.12063

Acknowledgments

The authors thank Emmanuel Laurenceau-Cornec for insightful comments on an earlier version of this manuscript and for assistance with data acquisition; they also thank the authors of the studies analyzed here for their efforts in collecting the data and for providing it for our analyses. GLB acknowledges funding from the Simons Foundation Postdoctoral Fellowship in Marine Microbial Ecology (645921). ELC was funded by a NERC standard research grant (NE/S000348/1) and an Imperial College Research Fellowship. BBC acknowledges support from the National Environmental Research Council (NE/R015953/1) and the Horizon 2020 Framework Program (820989). BBC has also received funding from the European Union's Horizon 2020 Research and Innovation Program under grant agreement No. 820989 (project COMFORT, our common future ocean in the Earth system—quantifying coupled cycles of carbon, oxygen, and nutrients for determining and achieving safe operating spaces with respect to tipping points). The work reflects only the authors' view; the European Commission and their executive agency are not responsible for any use that may be made of the information the work contains.



- Belcher, A., Iversen, M. H., Giering, S., Riou, V., Henson, S., Berline, L., et al. (2016a). Depth-resolved particle-associated microbial respiration in the northeast atlantic. *Biogeosciences*, 13(17), 4927–4943. https://doi.org/10.5194/bg-13-4927-2016
- Belcher, A., Iversen, M. H., Manno, C., Henson, S. A., Tarling, G. A., & Sanders, R. (2016b). The role of particle associated microbes in remineralization of fecal pellets in the upper mesopelagic of the Scotia Sea, Antarctica. *Limnology & Oceanography*, 61(3), 1049–1064. https://doi.org/10.1002/lno.10269
- Betancourt, M., Byrne, S., Livingstone, S., & Girolami, M. (2017). The geometric foundations of hamiltonian monte carlo. *Bernoulli*, 23(4A), 2257–2298. https://doi.org/10.3150/16-BEJ810
- Bopp, L., Aumont, O., Cadule, P., Alvain, S., & Gehlen, M. (2005). Response of diatoms distribution to global warming and potential implications: A global model study. *Geophysical Research Letters*, 32(19). https://doi.org/10.1029/2005GL023653
- Buesseler, K. O., & Boyd, P. W. (2009). Shedding light on processes that control particle export and flux attenuation in the twilight zone of the open ocean. *Limnology & Oceanography*, 54(4), 1210–1232. https://doi.org/10.4319/lo.2009.54.4.1210
- Bürkner, P.-C. (2017). brms: An R package for bayesian multilevel models using stan. Journal of Statistical Software, 80(1), 1–28. https:// doi.org/10.18637/jss.v080.i01
- Cael, B., & Bisson, K. (2018). Particle flux parameterizations: Quantitative and mechanistic similarities and differences. Frontiers in Marine Science, 5, 395. https://doi.org/10.3389/fmars.2018.00395
- Cael, B., Chase, A., & Boss, E. (2020). Information content of absorption spectra and implications for ocean color inversion. *Applied Optics*, 59(13), 3971–3984. https://doi.org/10.1364/AO.389189
- Cael, B., & Follows, M. J. (2016). On the temperature dependence of oceanic export efficiency. *Geophysical Research Letters*, 43(10), 5170–5175. https://doi.org/10.1002/2016GL068877
- Carder, K. L., Steward, R. G., & Betzer, P. R. (1982). In situ holographic measurements of the sizes and settling rates of oceanic particulates. Journal of Geophysical Research, 87(C8), 5681–5685. https://doi.org/10.1029/JC087iC08p05681
- Carpenter, B., Gelman, A., Hoffman, M. D., Lee, D., Goodrich, B., Betancourt, M., et al. (2017). Stan: A probabilistic programming language. *Journal of Statistical Software*, 76(1). https://doi.org/10.18637/jss.v076.i01
- Cavan, E. L., & Boyd, P. (2018). Effect of anthropogenic warming on microbial respiration and particulate organic carbon export rates in the sub-Antarctic Southern Ocean. Aquatic Microbial Ecology, 82(2), 111–127. https://doi.org/10.3354/ame01889
- Cavan, E. L., Giering, S., Wolff, G., Trimmer, M., & Sanders, R. (2018). Alternative particle formation pathways in the eastern tropical north pacific's biological carbon pump. *Journal of Geophysical Research: Biogeosciences*, 123(7), 2198–2211. https://doi.org/10.1029/2018JG004392
- Cavan, E. L., Henson, S. A., & Boyd, P. W. (2019). The sensitivity of subsurface microbes to ocean warming accentuates future declines in particulate carbon export. Frontiers in Ecology and Evolution, 6, 230. https://doi.org/10.3389/fevo.2018.00230
- Clift, R., Grace, J. R., & Weber, M. E. (2005). Bubbles, drops, and particles, Mineola, New York: Dover Publications.
- Collins, J. R., Edwards, B. R., Thamatrakoln, K., Ossolinski, J. E., DiTullio, G. R., Bidle, K. D., et al. (2015). The multiple fates of sinking particles in the north atlantic ocean. *Global Biogeochemical Cycles*, 29(9), 1471–1494. https://doi.org/10.1002/2014GB005037
- Deibel, D. (1990). Still-water sinking velocity of fecal material from the pelagic tunicate dolioletta gegenbauri. Marine ecology progress series. Oldendorf, 62(1), 55–60. https://doi.org/10.3354/meps062055
- DeVries, T., & Primeau, F. (2011). Dynamically and observationally constrained estimates of water-mass distributions and ages in the global ocean. *Journal of Physical Oceanography*, 41(12), 2381–2401. https://doi.org/10.1175/JPO-D-10-05011.1
- DeVries, T., Primeau, F., & Deutsch, C. (2012). The sequestration efficiency of the biological pump. *Geophysical Research Letters*, 39(13). https://doi.org/10.1029/2012GL051963
- Diercks, A.-R., & Asper, V. L. (1997). In situ settling speeds of marine snow aggregates below the mixed layer: Black sea and gulf of mexico. Deep Sea Research Part I, 44(3), 385–398. https://doi.org/10.1016/S0967-0637(96)00104-5
- Dunne, J. P., Gnanadesikan, A., Sarmiento, J. L., & Slater, R. D. (2010). Technical description of the prototype version (v0) of tracers of phytoplankton with allometric zooplankton (topaz) ocean biogeochemical model as used in the princeton ifmip model. *Biogeosciences*, 7(11), 3593. https://doi.org/10.5194/bg-7-3593-2010.
- Durkin, C. A., Estapa, M. L., & Buesseler, K. O. (2015). Observations of carbon export by small sinking particles in the upper mesopelagic. *Marine Chemistry*, 175, 72–81. https://doi.org/10.1016/j.marchem.2015.02.011
- Dutkiewicz, S., Hickman, A., Jahn, O., Gregg, W., Mouw, C., & Follows, M. (2015). Capturing optically important constituents and properties in a marine biogeochemical and ecosystem model. *Biogeosciences*, *12*(14), 4447–4481. https://doi.org/10.5194/bg-12-4447-2015
- Ebersbach, F., Trull, T. W., Davies, D. M., & Bray, S. G. (2011). Controls on mesopelagic particle fluxes in the sub-antarctic and polar frontal zones in the Southern Ocean south of australia in summer-perspectives from free-drifting sediment traps. *Deep Sea Research Part II*, 58(21–22), 2260–2276. https://doi.org/10.1016/j.dsr2.2011.05.025
- Engel, A., Abramson, L., Szlosek, J., Liu, Z., Stewart, G., Hirschberg, D., & Lee, C. (2009). Investigating the effect of ballasting by caco3 in emiliania huxleyi, ii: Decomposition of particulate organic matter. *Deep Sea Research Part II*, 56(18), 1408–1419. https://doi.org/10.1016/j.dsr2.2008.11.028
- Engel, A., & Schartau, M. (1999). Influence of transparent exopolymer particles (tep) on sinking velocity of nitzschia closterium aggregates. Marine Ecology Progress Series, 182, 69–76. https://doi.org/10.3354/meps182069
- Finkel, Z. V., Beardall, J., Flynn, K. J., Quigg, A., Rees, T. A. V., & Raven, J. A. (2010). Phytoplankton in a changing world: Cell size and elemental stoichiometry. *Journal of Plankton Research*, 32(1), 119–137. https://doi.org/10.1093/plankt/fbp098
- Gehlen, M., Bopp, L., Emprin, N., Aumont, O., Heinze, C., & Ragueneau, O. (2006). Reconciling surface ocean productivity, export fluxes and sediment composition in a global biogeochemical ocean model. *Biogeosciences Discussions*, 3(3), 803–836. https://doi.org/10.5194/bgd-3-803-2006
- Gelman, A., & Hill, J. (2006). Data analysis using regression and multilevel/hierarchical models, Cambridge, England: Cambridge University Press.
- Gibbs, R. J. (1985). Estuarine flocs: Their size, settling velocity and density. Journal of Geophysical Research, 90(C2), 3249–3251. https://doi.org/10.1029/JC090iC02p03249
- Giering, S. L., Cavan, E. L., Basedow, S. L., Briggs, N., Burd, A. B., Darroch, L. J., et al. (2020). Sinking organic particles in the ocean-flux estimates from in situ optical devices. *Frontiers in Marine Science*, *6*, 834. https://doi.org/10.3389/fmars.2019.00834
- Guidi, L., Jackson, G. A., Stemmann, L., Miquel, J. C., Picheral, M., & Gorsky, G. (2008). Relationship between particle size distribution and flux in the mesopelagic zone. *Deep Sea Research Part I*, 55(10), 1364–1374. https://doi.org/10.1016/j.dsr.2008.05.014
- Hawley, N. (1982). Settling velocity distribution of natural aggregates. Journal of Geophysical Research: Oceans, 87(C12), 9489–9498. https://doi.org/10.1029/JC087iC12p09489



Herndl, G. J., & Reinthaler, T. (2013). Microbial control of the dark end of the biological pump. *Nature Geoscience*, 6(9), 718–724. https://doi.org/10.1038/ngeo1921

- Hill, P. S., Syvitski, J. P., Cowan, E. A., & Powell, R. D. (1998). In situ observations of floc settling velocities in glacier bay, alaska. Marine Geology, 145(1–2), 85–94. https://doi.org/10.1016/S0025-3227(97)00109-6
- Iversen, M. H., Nowald, N., Ploug, H., Jackson, G. A., & Fischer, G. (2010). High resolution profiles of vertical particulate organic matter export off cape blanc, mauritania: Degradation processes and ballasting effects. *Deep Sea Research Part I*, 57(6), 771–784. https://doi. org/10.1016/j.dsr.2010.03.007
- Iversen, M. H., Pakhomov, E. A., Hunt, B. P., van der Jagt, H., Wolf-Gladrow, D., & Klaas, C. (2017). Sinkers or floaters? contribution from salp pellets to the export flux during a large bloom event in the Southern Ocean. *Deep Sea Research Part II: Topical Studies in Oceanog-raphy*, *138*, 116–125. https://doi.org/10.1016/j.dsr2.2016.12.004
- Iversen, M. H., & Ploug, H. (2010). Ballast minerals and the sinking carbon flux in the ocean: Carbon-specific respiration rates and sinking velocity of marine snow aggregates. *Biogeosciences*, 7(3), 2613–2624. https://doi.org/10.5194/bgd-7-3335-2010
- Iversen, M. H., & Ploug, H. (2013). Temperature effects on carbon-specific respiration rate and sinking velocity of diatom aggregates-potential implications for deep ocean export processes. *Biogeosciences*, 10(6), 4073. https://doi.org/10.5194/bg-10-4073-2013
- Iversen, M. H., & Robert, M. L. (2015). Ballasting effects of smectite on aggregate formation and export from a natural plankton community. Marine Chemistry, 175, 18–27. https://doi.org/10.1016/j.marchem.2015.04.009
- Jackson, G. A. (2015). Coagulation in a rotating cylinder. Limnology and Oceanography: Methods, 13(4), 194–201. https://doi.org/10.1002/ lom3.10018
- Johnson, C. P., Li, X., & Logan, B. E. (1996). Settling velocities of fractal aggregates. *Environmental Science & Technology*, 30(6), 1911–1918. https://doi.org/10.1021/es950604g
- Jouandet, M.-P., Trull, T. W., Guidi, L., Picheral, M., Ebersbach, F., Stemmann, L., & Blain, S. (2011). Optical imaging of mesopelagic particles indicates deep carbon flux beneath a natural iron-fertilized bloom in the Southern Ocean. *Limnology & Oceanography*, 56(3), 1130–1140. https://doi.org/10.4319/lo.2011.56.3.1130
- Kajihara, M. (1971). Settling velocity and porosity of large suspended particle. Journal of the Oceanographical Society of Japan, 27(4), 158–162. https://doi.org/10.1007/BF02109135
- Klaas, C., & Archer, D. E. (2002). Association of sinking organic matter with various types of mineral ballast in the deep sea: Implications for the rain ratio. *Global Biogeochemical Cycles*, *16*(4), 1–13. https://doi.org/10.1029/2001GB001765
- Kramer, S. J., & Siegel, D. A. (2019). How can phytoplankton pigments be best used to characterize surface ocean phytoplankton groups for ocean color remote sensing algorithms? *Journal of Geophysical Research: Oceans*, 124(11), 7557–7574. https://doi. org/10.1029/2019JC015604
- Kriest, I. (2002). Different parameterizations of marine snow in a 1d-model and their influence on representation of marine snow, nitrogen budget and sedimentation. *Deep Sea Research Part I*, 49(12), 2133–2162. https://doi.org/10.1016/S0967-0637(02)00127-9
- Kriest, I., & Oschlies, A. (2008). On the treatment of particulate organic matter sinking in large-scale models of marine biogeochemical cycles. *Biogeosciences*, 5(4), 55–72. https://doi.org/10.5194/bgd-4-3005-2007
- Kwon, E. Y., Primeau, F., & Sarmiento, J. L. (2009). The impact of remineralization depth on the air-sea carbon balance. *Nature Geoscience*, 2(9), 630–635. https://doi.org/10.1038/ngeo612
- Laurenceau-Cornec, E. C., Le Moigne, F. A., Gallinari, M., Moriceau, B., Toullec, J., Iversen, M. H., et al. (2019). New guidelines for the application of stokes' models to the sinking velocity of marine aggregates. *Limnology & Oceanography*, 65(6), 1264–1285. https://doi. org/10.1002/lno.11388
- Laurenceau-Cornec, E. C., Trull, T. W., Davies, D. M., Christina, L., & Blain, S. (2015). Phytoplankton morphology controls on marine snow sinking velocity. *Marine Ecology Progress Series*, 520, 35–56. https://doi.org/10.3354/meps11116
- Laws, E. A. (1975). The importance of respiration losses in controlling the size distribution of marine phytoplankton. *Ecology*, 56(2), 419–426. https://doi.org/10.2307/1934972
- Loth, E. (2008). Drag of non-spherical solid particles of regular and irregular shape. *Powder Technology*, 182(3), 342–353. https://doi.org/10.1016/j.powtec.2007.06.001
- McDonnell, A. M., & Buesseler, K. O. (2010). Variability in the average sinking velocity of marine particles. *Limnology & Oceanography*, 55(5), 2085–2096. https://doi.org/10.4319/lo.2010.55.5.2085
- Najjar, R. G., Jin, X., Louanchi, F., Aumont, O., Caldeira, K., Doney, S. C., et al. (2007). Impact of circulation on export production, dissolved organic matter, and dissolved oxygen in the ocean: Results from phase ii of the ocean carbon-cycle model intercomparison project (ocmip-2). *Global Biogeochemical Cycles*, 21(3). https://doi.org/10.1029/2006GB002857
- Niemeyer, D., Kriest, I., & Oschlies, A. (2019). The effect of marine aggregate parameterisations on global biogeochemical model performance. *Biogeosciences*, 16, 3095–3111. https://doi.org/10.5194/bg-16-3095-2019
- Nowald, N., Fischer, G., Ratmeyer, V., Iversen, M. H., Reuter, C., & Wefer, G. (2009). In-situ sinking speed measurements of marine snow aggregates acquired with a settling chamber mounted to the cherokee rov. Paper presented at *Oceans 2009-europe*, (pp. 1–6). https://doi.org/10.1109/OCEANSE.2009.5278186
- Picheral, M., Guidi, L., Stemmann, L., Karl, D. M., Iddaoud, G., & Gorsky, G. (2010). The underwater vision profiler 5: An advanced instrument for high spatial resolution studies of particle size spectra and zooplankton. *Limnology and Oceanography*, 8(9), 462–473. https:// doi.org/10.4319/lom.2010.8.462
- Ploug, H., & Jørgensen, B. B. (1999). A net-jet flow system for mass transfer and microsensor studies of sinking aggregates. Marine Ecology Progress Series, 176, 279–290. https://doi.org/10.3354/meps176279
- Reynolds, R. W., Smith, T. M., Liu, C., Chelton, D. B., Casey, K. S., & Schlax, M. G. (2007). Daily high-resolution-blended analyses for sea surface temperature. *Journal of Climate*, 20(22), 5473–5496. https://doi.org/10.1175/2007JCLI1824.1
- Riley, J., Sanders, R., Marsay, C., Le Moigne, F. A., Achterberg, E. P., & Poulton, A. J. (2012). The relative contribution of fast and slow sinking particles to ocean carbon export. *Global Biogeochemical Cycles*, 26(1). https://doi.org/10.1029/2011GB004085
- Sarmiento, J. L., & Gruber, N. (2006). Ocean biogeochemical dynamics, Princeton, NJ: Princeton University Press.
- Shanks, A. L., & Edmondson, E. (1989). Laboratory-made artificial marine snow: A biological model of the real thing. *Marine Biology*, 101(4), 463–470. https://doi.org/10.1007/BF00541648
- Shanks, A. L., & Trent, J. D. (1980). Marine snow: Sinking rates and potential role in vertical flux. Deep Sea Research Part A. Oceanographic Research Papers, 27(2), 137–143. https://doi.org/10.1016/0198-0149(80)90092-8
- Sigman, D. M., & Boyle, E. A. (2000). Glacial/interglacial variations in atmospheric carbon dioxide. *Nature*, 407(6806), 859–869. https://doi.org/10.1038/35038000



- Sigman, D. M., Hain, M. P., & Haug, G. H. (2010). The polar ocean and glacial cycles in atmospheric co 2 concentration. *Nature*, 466(7302), 47–55. https://doi.org/10.1038/nature09149
- Smayda, T. J. (1969). Some measurements of the sinking rate of fecal pellets. *Limnology & Oceanography*, 14(4), 621–625. https://doi. org/10.4319/lo.1969.14.4.0621
- Stokes, G. G. (1851). On the Effect of the Internal Friction of Fluids on the Motion of Pendulums. Cambridge: Pitt Press. https://doi. org/10.1017/CBO9780511702266.002
- Syvitski, J. P., Asprey, K., & Leblanc, K. (1995). In-situ characteristics of particles settling within a deep-water estuary. Deep Sea Research Part II, 42(1), 223–256. https://doi.org/10.1016/0967-0645(95)00013-G
- Tang, P., & Raper, J. A. (2002). Modeling the settling behavior of fractal aggregates-a review. Powder Technology, 123(2-3), 114–125. https://doi.org/10.1016/S0032-5910(01)00448-X
- Taucher, J., Bach, L. T., Riebesell, U., & Oschlies, A. (2014). The viscosity effect on marine particle flux: A climate relevant feedback mechanism. Global Biogeochemical Cycles, 28(4), 415–422. https://doi.org/10.1002/2013GB004728
- Van der Jagt, H., Friese, C., Stuut, J.-B. W., Fischer, G., & Iversen, M. H. (2018). The ballasting effect of saharan dust deposition on aggregate dynamics and carbon export: Aggregation, settling, and scavenging potential of marine snow. *Limnology & Oceanography*, 63(3), 1386–1394. https://doi.org/10.1002/lno.10779
- Völker, C., Laufkötter, C., Vogt, M., Gruber, N., Aumont, O., Bopp, L., et al. (2016). Projected decreases in future marine export production: The role of the carbon flux through the upper ocean ecosystem. *Biogeosciences*, 13(13), 4023–4047. https://doi.org/10.5194/ bg-13-4023-2016
- Williams, R. G., & Follows, M. J. (2011). Ocean dynamics and the carbon cycle: Principles and mechanisms, Cambridge, England: Cambridge University Press. https://doi.org/10.1017/CBO9780511977817
- Yool, A., Popova, E., & Anderson, T. (2013). Medusa-2.0: An intermediate complexity biogeochemical model of the marine carbon cycle for climate change and ocean acidification studies. *Geoscientific Model Development*, 6(5), 1767–1811. https://doi.org/10.5194/ gmdd-6-1259-2013
- Yoon, W., Kim, S., & Han, K. (2001). Morphology and sinking velocities of fecal pellets of copepod, molluscan, euphausiid, and salp taxa in the northeastern tropical atlantic. *Marine Biology*, 139(5), 923–928. https://doi.org/10.1016/S0025-3227(97)00109-6